Author's response to reviews

Title: Temperature differences are associated with malignancy on lung lesions: a clinical study

Authors:

Christodoulos Stefanadis (cstefan@cc.uoa.gr)
Christina Chrysohoou (chrysohoou@usa.net)
Dr Demosthenes Panagiotakos (d.b.panagiotakos@usa.net)
Elisabeth Passalidou (chrysohoou@usa.net)
Vailiki Katsi (chrysohoou@usa.net)
Vlassios Polychronopoulos (chrysohoou@usa.net)
Pavlos Toutouzas (chrysohoou@usa.net)

Version: 3 Date: 4 Jan 2003

Dear Editor

Thank you very much for the attention given to our work.
Please find below the formatting changes you have requested.

Formatting changes requested

Comment 1. * Title- Please remove all unnecessary capitalisation from the title
Reply 1. DONE

Comment 2. * Affiliations - Each institution should be written in full and be provided in the following format: Department, Institute, City, Country. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.
Reply 2. DONE

Comment 3. * Abstract - Abstracts must be structured into Background, Methods, Results, Conclusions. We ask that the background be written describing the context and purpose of the study.
Reply 3. DONE

Comment 4. * Manuscript - Manuscript sections should include (in the following order): Abstract, Background, Methods, Results, Discussion, Conclusions, Competing Interests, Authors' Contributions, Acknowledgements, References
* Table 1 - In the text, can we please ask you to refer to the table as Table 1
* Competing Interests - Please include. If there are none to declare, please write None declared.
* Author Contributorship - Please refer to the attached checklist for the preferred format
Reply 4. DONE

With our best wishes for the New Year
Sincerely
D B Panagiotakos